Author’s response to reviews

Title: Micro-morphologic changes around biophysically-stimulated titanium implants in ovariectomized rats

Authors:

Kivanc Akca (akcak@hacettepe.edu.tr)
Ebru Sarac (drebrusarac@yahoo.com)
Ugur Baysal (ubaysal@hacettepe.edu.tr)
Mete Fanuscu (mfanuscu@ucla.edu)
Ting Ling Chang (tlchang@dentnet.dent.ucla.edu)
Murat Cehreli (mccehreli@yahoo.com)

Version: 3 Date: 14 July 2007

Author’s response to reviews:

Dear Editor,

In accordance with the remarks of the reviewers, the following amendments were made in the manuscript:

Reviewer Dr. Wildemann:
1. The legend of Figure 3 is revised as Figure 3a and Figure 3b.

Reviewer Dr. Plate:
1. "Physiology" is corrected
2. A paper by Meyer et al. is already cited in the manuscript.
3. punctuation mark "x" corrected
4. "placed" deleted.
5. "after "histomorphometry" (Lit ? or document ?)": reference #26 is used.
7. miss: table 6 in the text?: included at the end of results section.

With kind regards,

Dr. Murat Cehreli